Author's response to reviews

Title: Interest of the modified Medical Research Council scale for the assessment of dyspnea in daily living in obesity: a pilot study

Authors:

Claire Launois (launois.claire@gmail.com)
Coralie Barbe (cbarbe@chu-reims.fr)
Eric Bertin (ebertin@chu-reims.fr)
Julie Nardi (jnardi@chu-reims.fr)
Jeanne-Marie Perotin (jperotin-collard@chu-reims.fr)
Sandra Dury (sdury@chu-reims.fr)
François Lebargy (flebargy@chu-reims.fr)
Gaetan Deslee (gdeslee@chu-reims.fr)

Version: 2 Date: 26 April 2012

Author's response to reviews:

Dear Editor,

Please find the modifications you required.

The study was reviewed and approved by the Institutional Review Board of the University Hospital of Reims, and patient consent was waived.

Please let me know if you need any other information.

Regards.

Dr Claire LAUNOIS